Author's response to reviews

Title: Congenital syphilis, still a reality in 21st century- a case report

Authors:

Monica Chaudhary (drmonica74@gmail.com)
Bineeta Kashyap (dr_bineetakashyap@yahoo.co.in)
Preena Bhalla (preenadr@gmail.com)

Version: 2 Date: 6 June 2007

Author's response to reviews:

As per the E-mail received from Mr. Joseph Crossland, Assistant Editor, we have included the conclusion section, competing interests section, authors contribution section and an acknowledgement section describing the informed consent from the patient.